Author's response to reviews

Title: Microvessel density as a new prognostic marker after radiotherapy in rectal cancer

Authors:

Saulius Svagzdys (s.svagzdys@gmail.com)
Vaiva Lesauskaite (vaiva.lesauskaite@med.kmu.lt)
Dainius Pavalkis (dainius.pavalkis@kmuk.lt)
Irena Nedzelskiene (irenaned@med.kmu.lt)
Darius Pranys (patkmuk@kmuk.lt)
Algimantas Tamelis (kp@kmuk.lt)

Version: 4 Date: 10 March 2009

Author's response to reviews: see over
Covering letter
Microvessel density as a new prognostic marker after radiotherapy in rectal cancer

Dear editorial board,
Thank you for assessing our manuscript once more. We have made corrections according to the advices presented by The BioMed Central Editorial Production Team. We would like to comment our additions.

We corrected authors list exactly as they are in the submission system:
Saulius Svagzdys¹§, Vaiva Lesauskaite²*, Dainius Pavalkis¹*, Irena Nedzelskiene⁴*, Darius Pranys³*, Algimantas Tamelis¹*

Verticals lines from tables are removed.
References list is checked and amended. We don’t use superscripts during citation.
“Reference List” is removed.
Figures are cropped as far as possible.

We send you corrected manuscript and figures once more.
Respectfully
Saulius Svagzdys and joint authors.